Author’s response to reviews

Title: Study on the Evaluation of the Clinical Effects of Traditional Chinese Medicine in Heart Failure by Complex Intervention: Protocol of SECETCM-HF

Authors:

Jingyuan Mao (jymao@126.com)
Yazhu Hou (mreleven@163.com)
Hongcai Shang (shanghongcai@126.com)
Henghe Wang (henghewang@126.com)
Xianliang Wang (xlwang1981@126.com)
Yingqiang Zhao (zhaoyingqiang1000@126.com)
Tianfu Niu (niutianfu2584@sina.com)
Jinrong Cui (cjinrong21417@sina.com)
Guangping Li (tjcardiol@126.com)
Qian Lin (lingqian@mail.cintcm.ac.cn)
Le Shi (xfile01@sina.com)
Xiuli Jia (xiulijia123@163.com)
Ruihong Fan (fanruihong6399@yahoo.com.cn)
Baoheng Wang (wb3423@sina.com)
Hongwu Wang (wanghw55@tjutcm.edu.cn)
Jishou Ruan (jsruan@nankai.edu.cn)

Version: 3 Date: 13 October 2009

Author’s response to reviews:

The objective of the trial is to establish the evaluation index system for the integration of TCM and western medicine. The evaluation indexes which include Traditional Chinese Medicine (TCM) items will specify the research design and methods. By putting the protocol into practice, the clinical effects of TCM for heart failure (HF) will be identified scientifically, objectively as well as rationally. The proper index system which built in the trial will be helpful for the clinical effect expression of HF treated with integrated medicine in future.

Each author has participated sufficiently in the work to take public responsibility for appropriate portions of the content. All authors read and approved the final manuscript and declare no competing interests.